Author’s response to reviews

Title: Meta-analysis comparing laparoscopic versus open resection for gastric gastrointestinal stromal tumors larger than 5 cm

Authors:

Xiao Lian (lxlx119@126.com)
Fan Feng (surgeonfengfan@163.com)
Man Guo (1246504365@qq.com)
Lei Cai (caileifmmu@qq.com)
Zhen Liu (doctorliuzhen@163.com)
Shangshu Liu (1437167914@qq.com)
Shuao Xiao (yydbddyx@sina.com)
Gaozan Zheng (136697876@qq.com)
Guanghui Xu (xuguanghui8@126.com)
Hongwei Zhang (zhanghwfmmu@126.com)

Version: 5 Date: 30 Oct 2017

Author’s response to reviews:

Editor Comments:

1. The individual contributions of ALL authors to the manuscript should be specified in the Authors’ Contributions section.
Answer: Thank you for your reminder. We have already provided clarification on all authors’ contributions to the manuscript in the Authors’ Contributions section.

2. Please include an Availability of Data and Materials statement in the Declarations section in one of the formats indicated in our submission guidelines.
Answer: Thank you for your reminder. We have added this section in the Declarations part.
3. Please include a "Consent for publication" statement in the "Declarations" section.

Answer: Thank you for your comments. We have added this section in the Declarations part.